UCSF UC San Francisco Previously Published Works

Title

Parent version of the Eating Disorder Examination: Reliability and validity in a treatment-seeking sample.

Permalink https://escholarship.org/uc/item/3wr218b8

Journal Journal of Eating Disorders, 12(1)

ISSN

2050-2974

Authors

Hail, Lisa Drury, Catherine McGrath, Robert <u>et al.</u>

Publication Date

2024-07-18

DOI

10.1186/s40337-024-01062-4

Peer reviewed

RESEARCH



Parent version of the Eating Disorder Examination: Reliability and validity in a treatment-seeking sample



Lisa Hail¹, Catherine R. Drury^{1,2*}, Robert E. McGrath², Stuart B. Murray³, Elizabeth K. Hughes^{4,5}, Susan M. Sawyer⁵, Daniel Le Grange^{1,6} and Katharine L. Loeb^{2,7}

Abstract

Background Assessment of eating disorders (ED) in youth relies heavily on self-report, yet persistent lack of recognition of the presence and/or seriousness of symptoms can be intrinsic to ED. This study examines the psychometric properties of a semi-structured interview, the parent version of the Eating Disorder Examination (PEDE), developed to systematically assess caregiver report of symptoms.

Methods A multi-site, clinical sample of youth (N = 522; age range: 12 to 18 years) seeking treatment for anorexia nervosa (AN) and subsyndromal AN were assessed using the Eating Disorder Examination (EDE) for youth and the PEDE for collateral caregiver report.

Results Internal consistencies of the four PEDE subscales were on par with established ranges for the EDE. Significant medium-sized correlations and poor to moderate levels of agreement were found between the corresponding subscales on each measure. For the PEDE, confirmatory factor analysis of the EDE four-factor model provided a poor fit; an exploratory factor analysis indicated that a 3-factor model better fits the PEDE.

Conclusions Findings suggest that the PEDE has psychometric properties on par with the original EDE. The addition of the caregiver perspective may provide incremental information that can aid in the assessment of AN in youth. Future research is warranted to establish psychometric properties of the PEDE in broader transdiagnostic ED samples.

Plain English summary

Assessments for eating disorders rely primarily on self-report; yet, the denial of symptoms or symptom severity among adolescents with anorexia nervosa can complicate assessment and delay treatment in this population. The Parent Eating Disorder Examination (PEDE) is the first semi-structured interview formally developed to improve childhood eating disorder assessment by including caregiver perspectives. In this study, a large sample of adolescents with anorexia nervosa completed a self-report interview (the Eating Disorder Examination or EDE) and their parents completed the PEDE. The PEDE appeared to measure parents' report of their child's eating disorder symptoms consistently. Results from both interviews were related to one another but did not completely agree.

*Correspondence: Catherine R. Drury catherine.drury@ucsf.edu

Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicate dot events in a credit line to the material. If material is not included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

This suggests that in an eating disorder assessment, the PEDE can provide additional information from caregivers that might reduce diagnostic confusion and lead to earlier intervention for youth with anorexia nervosa.

Keywords Eating disorder, Anorexia nervosa, Assessment, Adolescents, Parents

Background

With the publication of the fifth edition of the *Diagnostic and Statistical Manual of Mental Disorders* (*DSM-5*) [1], the criteria for diagnosing eating disorders (ED) were revised to reflect greater developmental sensitivity for youth. These modifications were particularly important as the onset of ED is most common in adolescence [2]. However, there remain many challenges to diagnosing restrictive ED, such as anorexia nervosa (AN), in children and adolescents, which could delay treatment of a pernicious, often refractory disorder.

One of the most significant challenges in relying on self-report in ED assessment and case identification is the persistent lack of recognition of the seriousness of symptoms, a core diagnostic feature of AN, which renders history of illness and present symptoms vulnerable to inaccuracies [3–5]. However, typical assessment methods for ED rely primarily on self-report, and may therefore be insufficient, particularly for younger individuals [6–9]. Compared to adults, adolescents generally score lower on measures of ED pathology despite similar levels of malnutrition [10], and appear to experience ED symptoms differently [11]. Minimization might be intrinsic to a developmentally normative limitation in recognizing the potential consequences of risky behaviors such as those associated with ED [5, 10, 12, 13]. Shorter duration of illness could compound this, further limiting adolescents' appreciation of the current and future impact of what could in fact become a severe and enduring disorder [14, 15]. Relatedly, adolescents are unlikely to independently seek help for their ED, and may even engage in strategic minimization of symptoms, to avoid the implications of symptom endorsement (e.g., intervention efforts on the part of adults).

In addition, there are cognitive and emotional obstacles to evaluating symptoms of AN in youth, as several of the criteria are psychological in nature. For example, the ability to report a fear of weight gain requires that the young person be able to recognize and label their affective state correctly, and to identify the motivation behind their behavior [15, 16]. Other criteria are more abstract in nature (e.g., disturbance in the experience of shape and weight, undue influence of shape and weight on selfevaluation), and require the developmental maturation of abstract reasoning to recognize and endorse ED symptoms [2, 17–19].

The utility of multi-informant methods of assessing child psychopathology is long-established, and approaches have advanced over time [6, 20, 21]. However, most measures used for youth with ED - with notable exceptions [3] – rely exclusively on direct patient report [7] despite the unique risks posed by false negatives in case identification, particularly of AN. Two studies have examined parent-child concordance on the Eating Disorder Examination (EDE) [22] by administering the interview to parents with minimal modifications to the measure [18, 23]. For example, Couturier and colleagues [23] simply changed wording of questions from you to your child and retained items reflecting the internal experience of the child without prompting parents for data on why and how these experiences can be inferred through the child's behavior. They found that youth with AN scored lower than their parents on two EDE subscales (Restraint and Weight Concern), while Mariano and colleagues [18] found good concordance between youth and parent scores. Mariano and colleagues [18] proposed that adolescents in their study were less likely to minimize their symptoms due to the timing of EDE administration (i.e., at the end of a two-day psychological assessment). It is also possible that more extensive adaptations to the EDE are needed to assist parents in consistently providing a comprehensive report of symptoms.

To address the need for a standardized method for including parental report in the assessment of ED, we developed a parent version of the EDE (PEDE) [7, 24, 36], with permission and input from the first author of the original measure [22], that mirrors the EDE but includes detailed questions to assess for observable indicators of ED. Although the EDE can be administered as young as 14 years and has been adapted for use in children aged 8 years and older [25], these assessments do not incorporate caregiver perspectives. Thus, the overall objective of the current study was to evaluate the psychometric properties of the PEDE in a large, multi-site sample of children and adolescents seeking treatment for AN and subsyndromal AN (SAN). Specifically, we examined the internal consistency of the PEDE subscales and the PEDE's convergent and construct validity in relation to the EDE. We also aimed to compare PEDE and EDE rates of AN diagnosis. We hypothesized that:

 Internal consistency of the four PEDE subscales (Restraint, Eating Concern, Weight Concern, Shape Concern) and global score as measured by Cronbach's alpha reliability coefficients would be similar to those previously established for the EDE, which found values in the range from .44 to .85 [26].

- 2. Convergent validity would be demonstrated by small, positive correlations and low to moderate levels of agreement between the established EDE subscales and the corresponding subscales on the PEDE. These positive relationships would indicate that the subscales are tapping into similar constructs. The small effect sizes and low to moderate levels of agreement (as opposed to good or excellent) would suggest that both youth and parent perspectives offer incremental information to an ED assessment [23].
- 3. Both the EDE and PEDE would have a different factor structure than the four original subscales presented by Cooper and colleagues [27] as no studies evaluating factor structure in the EDE have confirmed this model. No specific hypotheses were possible for the expected factor structure given variations in the results of three prior studies [28–30], only one of which included adolescents [28].
- 4. The PEDE would yield a diagnosis of AN more frequently than the EDE among participants with both AN and SAN.

Methods

Participants

Participants were youth and guardian informants who presented to two research-based ED treatment programs in the United States (US; New York and Chicago) and one in Melbourne, Australia. Researchers at these sites received training on the EDE and PEDE, administered both interviews to youth and their caregivers presenting to clinical research centers for treatment of a suspected ED, and contributed deidentified baseline data as part of this multisite collaboration to establish the PEDE's psychometric properties. Any larger studies [31, 32] from which these deidentified data were derived for secondary analyses were approved by the respective institutions' institutional review boards; the present study was designated exempt from board review.

In order to assess the reliability and validity of the PEDE in a relatively homogenous sample, this study focused specifically on youth presenting to these sites with probable AN or SAN [32], a site-specific research category that would fall under other specified feeding and eating disorder (OSFED) in *DSM-5-TR* nomenclature. The original inspiration for developing the PEDE was to help identify true caseness in the context of underweight ED where denial and minimization are prominent and therefore parental report may be most useful [23]. Thus, submitted cases (n=833) were excluded from analysis if one or more of the following were met: (a) percent expected body weight (EBW) based on median body mass index (mBMI) was greater than 100% (n=232, 27.85%), (b) criteria for bulimia nervosa or binge eating disorder might be met by virtue of 12 or greater EDE objective bulimic episodes in the past three months and weight>85% of EBW (n=0), (c) age was younger than 12 years (n=83, n=1)9.96% of the full sample), or (d) there was insufficient information to accurately determine EBW (n=1). Although low weight is a relative, personalized construct and population norms are not a valid benchmark against which to determine individual-level weight status, these weight criteria were used to reduce the likelihood of false positives and because not all sites recorded a more individualized measure of EBW and all reported percent of mBMI. The resulting sample included 522 youth paired with guardian informants, ranging in age from 12 to 18 years (M=15.4; SD=1.7), 89.7% parent- or self-identified as female, who were at 54–99% of mBMI (M=84.3%; SD=8.5). Further demographic data (including caregiver gender identity) were not reported consistently across all sites. The majority of participants were recruited from sites in Chicago (n=219; 42.0%) and Melbourne (n=260; 49.8%); 8.2% of participants (n=43) were recruited from the New York-based site. There was a significant difference in PEDE global scores across sites (F(2,6)=7.49), p=.002, $\eta^2=0.03$), with guardians in New York reporting higher levels of ED pathology than those in Chicago or Melbourne (p=.002). EDE global scores did not significantly differ across sites (p = .725).

Measures

Eating Disorder Examination (EDE) Version 16.0

The EDE [22] is a semi-structured clinical interview that was originally developed for use with adults but is also used, and has been found psychometrically acceptable, as a diagnostic and predictive tool with younger populations [33, 34]. The EDE is comprised of 33 items and uses a 7-point scale to measure the frequency (0 = "absence of")the feature"; 6 = "feature present every day") and severity (0 = "absence of the feature;" 6 = "feature present to an extreme degree") of ED attitudes and behaviors. Most of the questions capture data from the past 28 days only, with exception of the ten diagnostic items that extend to the previous three months to reflect the time frame evaluated to make the DSM ED diagnoses. The EDE includes four subscales: Restraint (5 items), Eating Concern (5 items), Shape Concern (8 items), and Weight Concern (5 items). The subscales are averaged to give a rating of global severity. Although these subscales have not been supported in a prior factor analysis, they remain widely used in both research and clinical practice [18].

Parent Eating Disorder Examination (PEDE)

The PEDE version 1.4 [24] includes items that directly mirror the content and 7-point scoring scheme of the EDE. While the term "parent" is used, this measure is appropriate to use with any adult who is in the primary

caretaking role. In the parent version, endorsement or denial (depending on the item) of a stem question triggers additional queries about behavioral observations and indicators of intent that are not present in the patientdirected EDE. Two additional items were added to the PEDE to assess for refusal to maintain a normal body weight and denial of the seriousness of low body weight, diagnostic features of AN that are not explicitly asked in the EDE. The item reaction to prescribed weighing from the EDE Weight Concern subscale was excluded because the item proved confusing when piloted. In total, the PEDE has 41 scored items. A symptom is rated as present if the parent has directly observed the phenomenon; heard the child report it; or heard reports from a reliable third party such as other family members, friends, or school personnel.

The PEDE requires that parents use their best judgment, including all available sources of information, in responding to the items. For example, in assessing *fear of weight gain*, there is not only an item evaluating verbal expression of this fear but also subsequent items assessing for indications that the young person is refusing attempts to increase their weight "by passive resistance (e.g., refusing to eat) and/or active resistance, such as yelling, throwing a tantrum, throwing food or dishes, running away, threatening to hurt themself if made to eat," or other means. Other examples include specific questions that evaluate evidence of purging behaviors (e.g., "Have you noticed any vomit residue or odor in the bathroom or on your child's clothes?; "Has your child rushed to the bathroom during a meal or immediately after eating?").

The PEDE version 1.4 was developed from the EDE version 16.0 [22] and contains diagnostic items consistent with *DSM-IV-TR* [35] diagnostic criteria. Additionally, the PEDE items that assess for behavioral indicators allow for the evaluation of the revised *DSM-5* criteria, including those criteria that are not explicitly assessed by the EDE version 16.0 or 17.0 (i.e., *refusal to maintain a normal body weight* and *denial of the seriousness of low body weight*). The PEDE version 2.0 has since been revised aligning the measure with *DSM-5* diagnostic criteria and incorporating gender-neutral language, and is publicly available [36].

Statistical Analyses

Cronbach's alpha coefficients were calculated to evaluate the internal consistency of the EDE and PEDE subscales and global scores using IBM SPSS Statistics v.24.0, with values less than .5 considered to be unacceptable, greater than or equal to .5 poor, greater than or equal to .6 questionable, greater than or equal to .7 acceptable, greater than or equal to .8 good, and greater than or equal to .9 excellent [37]. Convergent validity was assessed through the correlation and level of agreement between the EDE and PEDE subscales. Specifically, bivariate Pearson correlations were calculated using IBM SPSS Statistics v.24.0; as suggested by Cohen [38], .10 was considered a weak or small correlation, .30 medium, and .50 or larger strong or large. Additionally, the level of agreement between the EDE and PEDE subscales and global scores was measured using a two-way random effects model (absolute agreement, average measures) intraclass correlation coefficient (ICC). In accordance with the 95% confidence interval of the ICC estimate, values less than .50 were considered evidence of poor agreement, between .50 and .75 moderate agreement, between .75 and .90 good agreement, and greater than .90 excellent agreement [39].

To assess the goodness of fit of the original four-factor structure of the traditional EDE subscales developed by Fairburn and colleagues [27], confirmatory factor analysis (CFA) was conducted with Mplus (version 8.0) [40]. Model fit was evaluated using incremental fit tests of a "good fit" [41, 42], including the Tucker-Lewis index (TLI) \geq .90 and comparative fit index (CFI) \geq .90. Two absolute measures of fit were also used: the standard root mean square residual (SRMR) \leq .08 and root mean square error of approximation (RMSEA) \leq .10 (<.05 preferred). The same procedure was repeated with the PEDE. Given the results of the CFA, an exploratory factor analysis (EFA) was conducted using IBM SPSS Statistics v.24.0 to determine if an alternate model was a better fit for the PEDE.

Planned analyses for diagnostic agreement between the PEDE and EDE included chi-squared tests and Cohen's kappa to compare each measure's diagnostic items.

Results

Internal Consistency

The coefficient alpha values for the four established subscales and global score of the EDE in the present sample ranged from acceptable to excellent: .86 for the Restraint scale, .75 for Eating Concern, .93 for Shape Concern, .83 for Weight Concern, and .93 for the global score. While the PEDE reliability coefficients for the Shape Concern and Weight Concern subscales (.85 and .74, respectively) and the global score (.80) fell in the acceptable to good ranges, alpha coefficients were poor (.59) for the Restraint subscale and unacceptable (.44) for the Eating Concern subscale.

Construct validity

Table 1 shows the results of the Pearson correlations. There were significant medium-sized positive correlations between the corresponding subscales and global scores (all p values<.001) ranging from .36 to .49. In each case, the correlation with the corresponding scale

	EDE									
PEDE	Restraint	n	Eating Concern	n	Shape Concern	n	Weight Concern	n	Global Score	n
Restraint	.36*	369	.32*	369	.29*	369	.24*	368	.33*	369
Eating Concern	.37*	368	.38*	368	.33*	368	.29*	367	.37*	368
Shape Concern	.39*	359	.36*	359	.43*	359	.38*	358	.42*	359
Weight Concern	.39*	363	.35*	363	.42*	363	.42*	362	.43*	363
Global Score	.48*	368	.44*	368	.47*	368	.43*	367	.49*	368

Table 1 Summary of Pearson Correlations Between Original Subscales on the EDE and PEDE

Note. EDE=Eating Disorder Examination; PEDE=Parent Eating Disorder Examination

* p<.001, two-tailed

 Table 2
 Estimates of Inter-rater Agreement Between EDE and PEDE

Scale	n	ICC	р	95% Cl
Restraint	369	.50	<.001	[.38, .59]
Eating Concern	368	.49	<.001	[.33, .60]
Shape Concern	359	.56	<.001	[.43, .66]
Weight Concern	362	.57	<.001	[.46, .66]
Global Score	368	.61	<.001	[.52, .69]

Note. EDE=Eating Disorder Examination; PEDE=Parent Eating Disorder Examination; ICC=intraclass correlation coefficient; CI=confidence interval

of the other instrument was higher than that with any other scale. Estimates of inter-rater agreement between the EDE and PEDE subscale and global scores are shown in Table 2. There was moderate agreement between the PEDE and EDE global scores and the Restraint, Shape Concern, and Weight Concern subscale scores, and poor agreement between the Eating Concern subscales.

The CFA for the EDE four-factor model, based on established subscales, approached an acceptable fit after removing the *preoccupation with shape or weight* item from the Weight Concern factor because of a negative loading (see Table 3 for standardized factor loadings): CFI=.90, TLI=.88, RMSEA=.09, and SRMR=.05. The CFA of the four-factor model for the PEDE provided a poor fit to the data: CFI=.70, TLI=.66, RMSEA=.11 (*SE*=.10, .11); and SRMR=.10.

For the EFA, the scree plot, parallel analysis, and Velicer's minimum average partial (MAP) tests were conducted, with the latter two based on SPSS macros developed by O'connor [43]. All three tests supported retaining a three-factor model for the PEDE. Principal axis factoring (PAF) and promax rotation (power=4) were used to extract the three factors. Loadings above .30 were used as evidence of a meaningful relationship between an item and a factor [44]. These three factors accounted for 47.7% of the total variance of the items; see Table 3. One item, avoidance of eating, was not associated with any scale due to insufficient loading. Looking at the items within each factor, they could be labeled as affective preoccupation with shape, weight, and eating (10 items, $\alpha = 0.87$, 30.6% of total variance), *importance* of shape, weight, and restriction (7 items, $\alpha = 0.75$, 9.1% of total variance), and *discomfort with eating and body display* (4 items, α = 0.58, 8.0% of total variance).

Diagnostic Agreement

We initially planned to assess diagnostic agreement between the PEDE and EDE using chi-squared tests and Cohen's kappa to compare each measure's diagnostic items. However, of those participants who were not missing any EDE diagnostic items (n=361), only 237 had no missing PEDE diagnostic items. A t-test comparison of those with and without missing PEDE diagnostic items found that participants without missing PEDE data had significantly higher PEDE global scores (p = .002) and significantly lower BMIs (p=.013) than participants who were missing PEDE diagnostic items. As the patients who could be included in this analysis appeared to have a more severe ED presentation than the remainder of the sample, results from a PEDE-EDE diagnostic comparison would be difficult to interpret. This confound precluded our conducting the planned analyses to assess diagnostic agreement.

Discussion

To our knowledge, the PEDE is the first semi-structured interview formally developed with the aim to improve ED assessment in youth through the addition of caregiver perspectives, helping to reduce Type II error rate and under-identification of symptoms in youth with ED [45]. This study investigated the psychometric properties of the PEDE in a relatively large, international, multisite sample of families seeking treatment for AN and SAN. As predicted, the internal consistency of the PEDE was within the range of what has been published for the EDE (.44 to .85) [26], though lower than the EDE's reliability in this sample. Regarding convergent validity, effect sizes were larger than the expected small effect size based on the meta-analytic evidence for parent-child correlation for both internalizing (.26) and externalizing (.32) disorders [6]. However, the lack of strong concordance between the EDE and PEDE subscales indicates that the information captured by the PEDE is not redundant with the EDE. This finding suggests that information from parent informants complements diagnostic and clinical

Table 3 Factor Loadings of PEDE and EDE Items

	PEDE Revised Subscales (n=286)			EDE Original Subscales (n=448)					
Measure Item	F1	F2	F3	Restraint	Eating Concern	Shape Concern	Weight Concern		
Restraint									
Restraint over eating	.02	.43	.21	.89					
Avoidance of eating	.01	.14	.26	.55					
Empty stomach	.43	03	.07	.81					
Food avoidance	19	.56	.18	.70					
Dietary rules	18	.68	.03	.81					
Eating Concern									
Preoccupation with food	10	.54	.02		.71				
Fear loss of control	.39	.17	13		.64				
Social eating	03	.09	.44		.61				
Secret eating	19	.07	.31		.22				
Guilt about eating	.37	.24	.06		.79				
Shape Concern									
Dissatisfaction with shape	.46	.12	.38			.85			
Preoccupation with shape or weight †	.55	.23	16			.71			
Importance of shape	.24	.62	07			.68			
Fear of weight gain	.56	.29	.00			.81			
Discomfort seeing body	.32	04	.33			.82			
Discomfort about exposure	.08	06	.83			.76			
Feelings of fatness	.92	11	.02			.88			
Flat stomach	.77	08	.00			.74			
Weight Concern									
Dissatisfaction with weight	.86	15	09				.83		
Desire to lose weight	.84	17	02				.84		
Reaction to prescribed weighing							.52		
Importance of weight	.28	.54	11				.67		
Eigenvalue	6.42	1.92	1.69						
Percent of variance	3.56	9.13	8.04						
Cronbach's alpha	.87	.75	.58						
Correlations									
F1									
F2	.53								
F3	.41	.20							
Restraint									
Eating Concern				.88					
Weight Concern				.81	.86				
Shape Concern				.84	.90	.94			

Note. PEDE=Parent Eating Disorder Examination; EDE=Eating Disorder Examination. Loadings from PEDE pattern matrix and from EDE Mplus standardized factor loadings. F1=affective preoccupation with shape, weight, and eating; F2=importance of shape, weight, and restriction; F3=discomfort with eating and body display. Loadings in bold are >.30

⁺ Item is traditionally included in the Weight Concern subscale as well

information over and above that obtained by youth selfreport. Specifically, the behavioral indicators and examples provided by the PEDE appear to elicit diagnostically relevant information from parents that might otherwise remain unreported. In clinical practice, such questioning can also serve to educate parents that these behaviors and beliefs are part of the ED and thereby improve their capacity to clinically monitor and intervene to support their child's recovery. The EDE is used with four subscales, yet none of the three studies that have examined the factor structure has replicated the four-factor model [28–30]. In this sample, the original factor structure approached an acceptable fit with the youth self-report data, but only after removing the *preoccupation with weight and shape* item from the Weight Concern subscale. Given the inconsistency of factor analysis results across studies of the EDE [26], it was not surprising that another underlying structure of three subscales seemed to provide the best fit for the PEDE.

Although the PEDE has an empirically derived, threefactor structure, the original four-subscale model of the PEDE was found to measure constructs similar to those measured by the corresponding EDE scales, based on significant positive associations between corresponding subscales on the youth and parent interview. As such, it is reasonable to utilize the PEDE based on the four-subscale model to maintain consistency for research purposes. When using it for exclusively clinical purposes, the threefactor model may provide more meaningful constructs. Prior research also suggests that the EDE global score is a more useful measure of ED pathology than its subscales [46]; in light of the current study's internal consistency and construct validity results, the PEDE global score may also provide a more valid interpretation of its findings.

Limitations of this study include a predominantly female, treatment-seeking sample with specific criteria applied, including the use of population norms (i.e., %mBMI) to determine weight eligibility instead of individualized weight status based on historical growth patterns. These limitations constrained an understanding of how the PEDE interview may perform in more diverse, transdiagnostic (including atypical AN), and non-treatment-seeking samples. Resource limitations prevented duplicate assessments by multiple interviewers to establish inter-rater reliability or compare ratings from caregivers of different genders, but this is worthy of future study, as is test-retest reliability. Furthermore, missing data precluded completion of the diagnostic agreement analyses originally proposed by this study. Although the intent of developing the PEDE was to aid in the identification of AN/SAN, future research should aim to evaluate the measure's ability to distinguish between transdiagnostic ED cases and non-cases (i.e., criterion validity) as compared to the EDE using samples of adolescents with ED, subsyndromal ED, and no ED, and sensitivity and specificity analyses such as receiver-operator characteristic (ROC) curves. Additional work is also needed to more thoroughly assess the PEDE's validity and predictive power, including its relationships with other measures of ED and non-ED symptoms, other parent-report measures, clinician-assigned diagnosis, and clinical outcomes. Finally, by applying more sophisticated multiinformant statistical methods [21], future research could establish how clinicians and researchers systematically integrate potentially conflicting perspectives from youth and their caregivers,.

Conclusion

In summary, the use of parental informants is consistent with the approach to assessment of other areas of psychopathology in youth in which collateral informants frequently aid in the evaluation and diagnosis process [6, 20, 47]. The introduction of the PEDE allows for a standardized way to incorporate caregiver reports to aid in the assessment of AN, potentially reducing diagnostic ambiguity and compensating for the denial and minimization inherent in the self-report of symptoms within the group. Our future research will focus on differences in diagnostic rates when parents are enlisted as informants in interview-based AN case identification efforts. Enhanced assessment approaches can theoretically make identification of clinically significant presentations more efficient and accurate, and lead to earlier intervention and improved outcomes.

Abbreviations

BMI	Body mass index
CFA	Confirmatory factor analysis
CFI	Comparative fit index
DSM	Diagnostic and Statistical Manual of Mental Disorders
EBW	Expected body weight
ED	Eating disorders
EDE	Eating Disorder Examination
EFA	Exploratory factor analysis
ICC	Intraclass correlation coefficient
MAP	Minimum average partial
mBMI	Median body mass index
OSFED	Other specified feeding and eating disorder
PAF	Principal axis factoring
PEDE	Parent Eating Disorder Examination
RMSEA	Root mean square error of approximation
SAN	Subsyndromal anorexia nervosa
SRMR	Standard root mean square residual
TLI	Tucker-Lewis index
US	United States

Acknowledgements

We gratefully acknowledge Christopher Fairburn's ongoing mentorship and support in forwarding research-based adaptations of the Eating Disorder Examination, including the parent version discussed in this paper.

Author contributions

KLL developed the PEDE and conceptualized the current study with LH, REM, and SBM. EKH, SMS, DLG, and KLL provided data for the study, which was curated and analyzed by LH under the supervision of REM and KLL. CRD contributed additional analyses with guidance from KLL and LH. LH wrote the initial draft of the manuscript, to which CRD, KLL, and REM also contributed. All authors read and edited subsequent iterations of the manuscript and approved the final version.

Funding

This research was supported by a grant from the National Institute of Mental Health K23 MH074506 (PI: Loeb; ClinicalTrials.gov NCT00418977, Early Identification and Treatment of Anorexia Nervosa).

Data availability

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request. The PEDE 2.0 is available at https://ccebt.com/wp-content/uploads/2024/06/PEDE-2.0_gender-neutral. pdf .

Declarations

Ethics approval and consent to participate

This study was determined to qualify as exempt by Fairleigh Dickinson University's Institutional Review Board (IRB). Any parent studies from which data were derived for secondary analyses included informed consent/assent and were approved by site-specific IRBs.

Consent for publication

Not applicable.

Competing interests

KLL receives royalties from Cambridge University Press and Routledge, and is a faculty member of and consultant for the Training Institute for Child and Adolescent Eating Disorders. DLG receives royalties from Guilford Press and Routledge, and is co-director of the Training Institute for Child and Adolescent Eating Disorders, LLC. SBM receives royalties from Oxford University Press, Routledge, and Springer.

Author details

¹Department of Psychiatry and Behavioral Sciences, University of California, San Francisco, UCSF Weill Institute for Neurosciences, 675 18th Street, San Francisco, CA, USA

²School of Psychology and Counseling, Fairleigh Dickinson University, Teaneck, NJ, USA

³Department of Psychiatry and the Behavioral Sciences, University of Southern California, Los Angeles, CA, USA

⁴Department of Paediatrics, The University of Melbourne, Melbourne, Australia

⁵Murdoch Children's Research Institute, Melbourne, Australia ⁶Department of Psychiatry and Behavioral Neuroscience (emeritus), The University of Chicago, Chicago, IL, USA

⁷Chicago Center for Evidence-Based Treatment, Chicago, IL, USA

Received: 29 March 2024 / Accepted: 10 July 2024 Published online: 18 July 2024

References

- 1. American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 5th ed. Washington: American Psychiatric Association; 2013.
- Lock J. An update on evidence-based psychosocial treatments for eating disorders in children and adolescents. J Clin Child Adolesc Psychol. 2015;44:707–21.
- Accurso EC, Waller G. Concordance between youth and caregiver report of eating disorder psychopathology: development and psychometric properties of the eating Disorder-15 for Parents/Caregivers (ED-15-P). Int J Eat Disord. 2021;54:1302–6.
- Vandereycken W. Denial of illness in anorexia nervosa—a conceptual review: part 1 diagnostic significance and assessment. Eur Eat Disord Rev. 2006;14:341–51.
- Vitousek KB, Daly J, Heiser C. Reconstructing the internal world of the eatingdisordered individual: overcoming denial and distortion in self-report. Int J Eat Disord. 1991;10:647–66.
- De Los Reyes A, Augenstein TM, Wang M, Thomas SA, Drabick DAG, Burgers DE, et al. The validity of the multi-informant approach to assessing child and adolescent mental health. Psychol Bul. 2015;141:858–900.
- Loeb KL, Brown M, Munk Goldstein M. Assessment of eating disorders in children and adolescents. In: Le Grange D, Lock J, editors. Eating disorders in children and adolescents: a clinical handbook. New York: Guilford Press; 2011. pp. 156–98.
- O'Logbon J, Newlove-Delgado T, McManus S, Mathews F, Hill S, Sadler K, et al. How does the increase in eating difficulties according to the Development and Well-Being Assessment screening items relate to the population prevalence of eating disorders? An analysis of the 2017 Mental Health in Children and Young people survey. Int J Eat Disord. 2022;55:1777–87.
- Swanson SA, Aloisio KM, Horton NJ, Sonneville KR, Crosby RD, Eddy KT, et al. Assessing eating disorder symptoms in adolescence: is there a role for multiple informants? Int J Eat Disord. 2014;47:475–82.
- 10. Couturier JL, Lock J. Denial and minimization in adolescents with anorexia nervosa. Int J Eat Disord. 2006;39:212–6.
- 11. Micali N, House J. Assessment measures for child and adolescent eating disorders: a review. Child Adolesc Ment Health. 2011;16:122–7.
- 12. Becker AE, Eddy KT, Perloe A. Clarifying criteria for cognitive signs and symptoms for eating disorders in DSM-V. Int J Eat Disord. 2009;42:611–9.
- Loeb KL, Jones J, Roberto CA, Sonia Gugga S, Marcus SM, Attia E, et al. Adolescent–adult discrepancies on the eating disorder examination: a function of developmental stage or severity of illness? Int J Eat Disord. 2011;44:567–72.

- Austin A, Flynn M, Richards K, Hodsoll J, Duarte TA, Robinson P, et al. Duration of untreated eating disorder and relationship to outcomes: a systematic review of the literature. Eur Eat Disord Rev. 2021;29:329–45.
- Fisher M, Schneider M, Burns J, Symons H, Mandel FS. Differences between adolescents and young adults at presentation to an eating disorders program. J Adolesc Health. 2001;28:222–7.
- 16. Cooper PJ, Watkins B, Bryant-Waugh R, Lask B. The nosological status of early onset anorexia nervosa. Psychol Med. 2002;32:873–80.
- Bravender T, Bryant-Waugh R, Herzog D, Katzman D, Kriepe RD, Lask B, et al. Classification of eating disturbance in children and adolescents: proposed changes for the DSM-V. Eur Eat Disord Rev. 2010;18:79–89.
- Mariano P, Watson HJ, Leach DJ, McCormack J, Forbes DA. Parent–child concordance in reporting of child eating disorder pathology as assessed by the eating disorder examination. Int J Eat Disord. 2013;46:617–25.
- Rosso IM, Young AD, Femia LA, Yurgelun-Todd DA. Cognitive and emotional components of frontal lobe functioning in childhood and adolescence. Ann NY Acad Sci. 2004;1021:355–62.
- Kuhn C, Aebi M, Jakobsen H, Banaschewski T, Poustka L, Grimmer Y, et al. Effective mental health screening in adolescents: should we collect data from youth, parents or both? Child Psychiatry Hum Dev. 2017;48:385–92.
- Martel MM, Markon K, Smith GT. Research review: multi-informant integration in child and adolescent psychopathology diagnosis. J Child Psychol Psychiatry. 2017;58:116–28.
- Fairburn CG, Cooper Z, O'Connor M. Eating disorder examination (16.0D). Fairburn CG. Cognitive behavior therapy and eating disorders. New York: Guilford Press; 2008. pp. 270–306.
- Couturier J, Lock J, Forsberg S, Vanderheyden D, Yen HL. The addition of a parent and clinician component to the eating disorder examination for children and adolescents. Int J Eat Disord. 2007;40:472–5.
- Loeb KL. Eating Disorder Examination Parent Version (P-EDE), version 1.4.
 2008. Unpublished measure based on Fairburn CG, Cooper Z, O'Connor M.
 Eating Disorder Examination (16.0D). In: Fairburn CG. Cognitive behavior therapy and eating disorders. New York: Guilford Press; 2008. pp. 270–306.
- Bryant-Waugh RJ, Cooper PJ, Taylor CL, Lask BD. The use of the eating disorder examination with children: a pilot study. Int J Eat Disord. 1996;19:391–7.
- Berg KC, Peterson CB, Frazier P, Crow SJ. Psychometric evaluation of the eating disorder examination and eating disorder Examination-Questionnaire: a systematic review of the literature. Int J Eat Disord. 2012;45:428–38.
- 27. Cooper Z, Cooper PJ, Fairburn CG. The validity of the eating disorder examination and its subscales. Br J Psychiatry. 1989;154:807–12.
- Byrne SM, Allen KL, Lampard AM, Dove ER, Fursland A. The factor structure of the eating disorder examination in clinical and community samples. Int J Eat Disord. 2010;43:260–5.
- Mannucci E, Ricca V, Di Bernardo M, Moretti S, Cabras PL, Rotella CM. Psychometric properties of EDE 12.0D in obese adult patients without binge eating disorder. Eat Weight Disord. 1997;2:144–9.
- Grilo CM, Crosby RD, Peterson CB, Masheb RM, White MA, Crow SJ, et al. Factor structure of the eating disorder examination interview in patients with binge-eating disorder. Obesity. 2010;18:977–81.
- Hughes EK, Le Grange D, Court A, Yeo MS, Campbell S, Allan E, et al. Parentfocused treatment for adolescent anorexia nervosa: a study protocol of a randomised controlled trial. BMC Psychiatry. 2014;14:105.
- Loeb KL, Weissman RS, Marcus S, Pattanayak C, Hail L, Kung KC, et al. Family-based treatment for anorexia nervosa symptoms in high-risk youth: a partially-randomized preference-design study. Front Psychiatry. 2020;10:985.
- Passi VA, Bryson SW, Lock J. Assessment of eating disorders in adolescents with anorexia nervosa: self-report questionnaire versus interview. Int J Eat Disord. 2003;33:45–54.
- Wade TD, Byrne S, Bryant-Waugh R. The eating disorder examination: norms and construct validity with young and middle adolescent girls. Int J Eat Disord. 2008;41:551–8.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 4th ed., text revision. Washington: American Psychiatric Association; 2000.
- Loeb KL. Eating Disorder Examination Parent Version (PEDE), Version 2.0. 2017. https://ccebt.com/wp-content/uploads/2024/06/PEDE-2.0_genderneutral.pdf. Accessed 16 June 2024.
- George D, Mallery P. SPSS for windows step by step: a simple guide and reference, 11.0 update. 4th ed. Boston: Allyn & Bacon; 2003.
- Cohen J. Statistical power analysis for the behavioral sciences. 2nd ed. New York: Routledge; 2013.

- 39. Koo TK, Li MY. A guideline of selecting and reporting intraclass correlation coefficients for reliability research. J Chiropr Med. 2016;15:155–63.
- Muthén LK, Muthén BO. Mplus user's guide. 7th ed. Los Angeles: Muthén & Muthén; 2015.
- 41. Byrne B. Structural equation modeling with LISREL, PRELIS, and SIMPLIS. Hillsdale: Lawrence Erlbaum; 1998.
- 42. Geiser C. Data analysis with Mplus. New York: Guildford; 2010.
- 43. O'connor BP. SPSS and SAS programs for determining the number of components using parallel analysis and Velicer's MAP test. Behav Res Methods Instrum Compu. 2000;32:396–402.
- 44. Costello AB, Osborne JW. Best practices in exploratory factor analysis: four recommendations for getting the most from your analysis. Pract Assess Res Eval. 2005;10:1–9.
- Murray SB, Loeb KL, Le Grange D. Indexing psychopathology throughout family-based treatment for adolescent anorexia nervosa: are we on track? Adv Eat Disord. 2014;2:93–6.

- Jenkins PE, Rienecke RD. Structural validity of the eating disorder Examination-Questionnaire: a systematic review. Int J Eat Disord. 2022;55(8):1012–30.
- Kraemer HC, Measelle JR, Ablow JC, Essex MJ, Boyce WT, Kupfer DJ. A new approach to integrating data from multiple informants in psychiatric assessment and research: mixing and matching contexts and perspectives. Am J Psychiatry. 2003;160:1566–77.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.